Author's response to reviews

Title: Predicting the outcome of chronic kidney disease by the estimated nephron number: The rationale and design of PRONEP, a prospective, multicenter, observational cohort study

Authors:

Toshiyuki Imasawa (imasawa@cehpnet.com)
Takashi Nakazato (tnakazato-circ@umin.ac.jp)
Hiroo Ikehira (i.hiroo@cehpnet.com)
Hiroyuki Fujikawa (f.hiroyu@cehpnet.com)
Ryo Nakajima (n.ryo@cehpnet.com)
Takahito Ito (i-taka@onh.go.jp)
Yutaka Ando (yutaando@ommc-hp.jp)
Mitsuhiro Yoshimura (yosimura@kinbyou.hosp.go.jp)
Masaru Nakayama (mnaka@kyumed.jp)
Kensei Yahata (kyahata@kyotolan.hosp.go.jp)
Osamu Sasaki (osasaki@nmh.hosp.go.jp)
Takao Yaomura (yaomurat@nnh.hosp.go.jp)
Ritsuko Katafuchi (katafuchir@fukuoka.e2.hosp.go.jp)
Tsuyoshi Yamamura (babiymama244@hok-mc.hosp.go.jp)
Takehiko Kawaguchi (k.takehiko@cehpnet.com)
Motonobu Nishimura (mnishi02@cehpnet.com)
Hiroshi Kitamura (hkitamu@cehpnet.com)
Takashi Kenmochi (kenmochi@cehpnet.com)
Akira Shimatsu (ashimats@kyotolan.hosp.go.jp)

Version: 2 Date: 6 February 2012

Author's response to reviews:

Executive Editor
BioMed Central

Dear Dr. Hayley Henderson

Now, I finished to rebuild my manuscript according to your requests. I will be so please that I have an opportunity to publish your journal.

Sincerly yours,
Toshiyuki Imasawa